Author's response to reviews

Title: Isolated angioedema of the bowel due to C1 esterase inhibitor deficiency: A case report and review of literature

Authors:

SHIVANGI T KOTHARI (shivangitkothari@gmail.com)
ROBERT SPIRA (cutedoc9@yahoo.com)
DEVIPRASAD BOTU (dpbotu@yahoo.com)
ROBERT GREENBLATT (itskots@rediffmail.com)
JOSEPH DEPASQUALE (joseph.depasquale@rediffmail.com)
ANISH M SHAH (anishshah21@gmail.com)

Version: 5 Date: 22 August 2010

Author's response to reviews: see over
Major revisions (we require the authors to make these changes)

Author list - The list of authors in the manuscript should be written exactly as they are in the submission system. The authors (forename, initial (if any), surname) should be written identically and in the order in which they are listed in the submission system.

Changes made accordingly.

Affiliations - Each affiliation should be written in full, in the following format: department, institute, city, and country. This may mean repeating institutional addresses if the authors are from different departments within an institution.

Changes made accordingly.

Patient information: please provide the ethnicity of the patient; 'Polish' is their nationality'.

Changes made accordingly.

Abstract: please do not use abbreviations in the abstract section.

Changes made accordingly.

Figure legends: please ensure that all figures are cited at least once; currently figures 2 and 3 have not been cited.

Changes made accordingly.

References - References should be cited in the text using consecutive numbers in square brackets. Please do not use superscript numbers. Also, please place the reference before the full stop. For example `text [1].' For example, reference 9 is cited before references 2-8, please amend this.

Changes made accordingly.

Minor revisions (We can make these changes on resubmission, although it will speed up publication of your manuscript if you do them while making the major changes above).

Kindly make the changes.

Figure cropping - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for
submission and can be viewed here:
http://www.biomedcentral.com/info/ifora/figures

Kindly make the changes.

Table format: please edit these so that vertical lines are not visible

Kindly make the changes.